Author's response to reviews

Title: Internet-based self-help treatment for depression in multiple sclerosis: study protocol of a randomized controlled trial

Authors:

Rosa E Boeschoten (r.boeschoten@ggzingeest.nl)
Joost Dekker (j.dekker@vumc.nl)
Bernard M J Uitdehaag (bmi.uitdehaag@vumc.nl)
Chris H Polman (ch.polman@vumc.nl)
Emma H Collette (eh.collette@vumc.nl)
Pim Cuijpers (p.cuijpers@vu.nl)
Aartjan T F Beekman (a.beekman@ggzingeest.nl)
Patricia van Oppen (p.vanoppen@ggzingeest.nl)

Version: 2 Date: 5 July 2012

Author's response to reviews:

5 july 2012

Dear sir/madam,

In reaction on your email which was sent on june 7, 2012; I uploaded the English and Dutch version of the ethical approval and approval of external funding for our study: 'Internet-based self-help treatment for depression in multiple sclerosis: study protocol of a randomized controlled trial'.

There are no manuscripts under submission with other journals based on this study protocol.

If there are any questions, please let me know.

Best wishes,

Rosa Boeschoten